Author's response to reviews

Title: Community-acquired K.pneumoniae meningitis in an alcoholic patient with an infected pancreatic pseudocyst; a case report and review of literature

Authors:

Adriana A Orzechowska Dr (adrianaorzechowska@hotmail.com)
Mark M Melzer Dr (mark.melzer@bhrhospitals.nhs.uk)
Sandra S Lacey Dr (sandra.lacey@bhrhospitals.nhs.uk)
Geraldine G Soosay Dr (geraldine.soosay@bhrhospitals.nhs.uk)

Version: 2 Date: 23 August 2007

Author's response to reviews:

Written consent from the patients widow has now been obtained.